Author’s response to reviews

Title: Comparison of C-MAC D-Blade videolaryngoscope and McCoy laryngoscope efficacy for nasotracheal intubation in simulated cervical spinal injury: a prospective randomized comparative study

Authors:
Kwon Seo (julinakh@hanmail.net)
Kyung Mi Kim (sumsonyo@gmail.com; sumsonyo@hallym.or.kr)
Hyunji John (hyunjiji@hallym.or.kr)
Joo Jun (ilpleut@hallym.or.kr)
Minsoo Han (msh805@hallym.or.kr)
Soyoun Kim (190384@hallym.or.kr)

Version: 2 Date: 21 Apr 2020

Author’s response to reviews:

April 20th, 2020
Editor of BMC anesthesiology

Dear D John Doyle, MD PhD

We would like to thank you and the reviewers of BMC anesthesiology for taking the time to review our article. We have made some corrections and clarification in the manuscript after going over the editor comments. The responses to the editor comments have been prepared and attached herewith.

&l<Responses to the editor comments&gt;
Editor Comments:

1. CONSORT

-- Please include a statement within your methods section to indicate that your study adheres to CONSORT guidelines.
⇒ We have added the statement within methods section (page 6, the first paragraph).
→ The present study adhered to CONSORT guidelines.

2. Consent for publication

-- Please amend your “Consent for publication” section in your Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.
We have revised the statement about that concern in consent for publication section as editor suggested.

Consent for publication: Not Applicable

3. Clean copy

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

We have revised the manuscript as a single, final, clean version as editor comments.

We hope that the revised manuscript will better meet the requirements of your journal for publication. We thank the editor and the reviewers of BMC anesthesiology once again for the constructive review of our paper. It is difficult time for everyone due to COVID-19. We hope you are staying safe and healthy through these unusual times!

Sincerely yours,

Kyung Mi Kim, M.D., Ph.D.
Assistant Professor
Department of anesthesiology and pain medicine
Hallym University Sacred Heart Hospital, Hallym University School of Medicine, 22, Gwanpyeong-ro 170 beon-gil, Dong-gu, Anyang-si, Gyeonggi-do, 14068, Republic of Korea.
Fax: 82-031-385-3244 Tel: 82-031-380-5959
E-mail: sumsonyo@gmail.com